Author's response to reviews

Title: Identifying structures, processes, resources and needs of research ethics committees in Egypt

Authors:

Hany Sleem (hanysleem@hotmail.com)
Samer S El-Kamary (selkamar@epi.umaryland.edu)
Henry J Silverman (hsilverm@medicine.umaryland.edu)

Version: 3 Date: 26 June 2010

Author's response to reviews:

We have made formatting changes, as requested by BMC Medical Ethics. We have also corrected several typographical errors and grammatical errors; we have also made corrections ensuring that the data in the manuscript matched what are reported in the tables.

Henry Silverman